Author's response to reviews

Title: A presenting case of disseminated MRSA infection. Is Daptomycin effective in the Central Nervous System infections? A case report.

Authors:

Fabrizio Taglietti (taglietti.f@gmail.com)
Floriana Campanile (campanile@unict.it)
Alessandro Capone (alessandro.capone@inmi.it)
Antonino Di Caro (antonino.dicaro@inmi.it)
Elisabetta Grilli (elisabetta.grilli@inmi.it)
Giulia Stazi (giulia.stazi@inmi.it)
Taschia Bertuccio (bertuccio@unict.it)
Nicola Petrosillo (nicola.petrosillo@inmi.it)
Stefania Stefani (stefanis@unict.it)

Version: 5 Date: 24 May 2012

Author's response to reviews: see over
To *Journal of Medical Case Reports*

The Editor

RE: manuscript submission.

May 24, 2012

Dear Editor,

These are the corrections requested.

1. references to support the statement: "Very few cases of infections due to CA-MRSA are described in our country: added (reference n° 7)

2. “Discussion and references supporting daptomycin use and efficacy are needed”: added (reference n° 14)

Referee n° 2:

1. “It should be mentioned this is a case of health-care associated S. aureus infection rather than a community-acquired one”: the MRSA phenotypic and genotypic characteristics suggest that this was a CA-MRSA infection, suggesting that the patient was colonized before the admission to the emergency room.

2. “The authors should make it clear whether any cerebral abscesses were drained or only treated by medical therapy”: correction line 53 in case presentation (highlighted)

3. Point 3: the problem is discussed in the discussion (highlighted in yellow).

4. We reviewed the English language.
Kind regards

**Corresponding author:**

**Fabrizio Taglietti, M.D.**

2nd Infectious Diseases Division,

National Institute for Infectious Diseases “L. Spallanzani”,

Via Portuense, 292-00149 Rome, Italy.

e-mail: [fabrizio.taglietti@inmi.it](mailto:fabrizio.taglietti@inmi.it)

tel +390655170241

fax +390655170315